Author's response to reviews

Title: A cost-effectiveness analysis of a preventive exercise program for patients with advanced head and neck cancer treated with concomitant chemo-radiotherapy

Authors:

Valesca P Retèl (v.retel@nki.nl)
Lisette van der Molen (l.vd.molen@nki.nl)
Frans JM Hilgers (f.hilgers@nki.nl)
Coen RN Rasch (c.rasch@nki.nl)
Annemiek AAMHJ l'Ortye (a.lortye@nki.nl)
Lotte MG Steuten (l.m.g.steuten@utwente.nl)
Wim H van Harten (w.v.harten@nki.nl)

Version: 5 Date: 2 November 2011

Author's response to reviews: see over
Formatting changes to be made:
---------------------------------

Major revisions (we require the author to make these changes)

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].`
Response: *We changed this*

Figures: You have included the figures in the manuscript. Please remove these.
Response: *We removed them*

Additional files: Please remove from the manuscript.
Response: *We removed them*

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove from the manuscript.
Response: *We removed them*

Author list: Please remove titles.
Response: *We removed them*

Structure: Change material and methods to methods.
Response: *We changed this*

Comments: Please remove post-it style comments. Eg: Insert table 1 here.
Response: *We removed this*

Please remove table used for list of abbreviations.
Response: *We removed this*

Tables: Please remove vertical lines.
Response: *We removed this*

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:
http://www.biomedcentral.com/info/ifora/figures
Response: *We checked this*

Figures: The image file should not include the title (e.g. Figure 1... etc.) or figure number.
Response: *We removed this*

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: *We checked this*